Author’s response to reviews

Title: Nearly half of Ultrio Plus NAT non-discriminated reactive blood donors were identified as occult HBV infection in south China

Authors:
Xianlin Ye (yexianlin90@hotmail.com)
Tong Li (dyltlt@163.com)
Wen Shao (2874732483@qq.com)
Jinfeng Zeng (zzengjf@163.com)
Wenxu Hong (wenxuhong@hotmail.com)
Liang Lu (339271353@qq.com)
Weigang Zhu (463229136@qq.com)
Chengyao Li (chengyaoli@hotmail.com)
Tingting Li (apple-ting-007@163.com)

Version: 3 Date: 05 Jun 2019

Author’s response to reviews:

Dear Dr. Oliver Cocks:

Ref: manuscript INFD-D-19-00121 entitled “Nearly half of Ultrio Plus NAT non-discriminated reactive blood donors were identified as occult HBV infection in south China”.

We would like to re-submit the new version of manuscript before 11 Jun 2019 as the journal required. According to your comments, the manuscript was carefully revised, including the full name of the ethics committee, consent for publication, the role of the funding body, the authors with the same initials, in a final, clean version of manuscript.

If you have any more question regarding to this manuscript, please let me know.

Yours sincerely,

Dr. Tingting Li